Author's response to reviews

Title: Extracorporeal shock wave therapy in the treatment of primary bone marrow edema syndrome of the knee: a prospective randomised controlled study

Authors:

Fuqiang Gao MD (gaofuqiang0604@163.com)
Wei Sun MD (18901267995@163.com)
Zirong Li MD (lizirong0297@163.com)
Wanshou Guo MD (guowanshou0927@163.com)
Weiguo Wang MD (wangweiguo421@163.com)
Liming Cheng MD (chengliming212@163.com)
Debo Yue MD (yuedebo22@163.com)
Nianfei Zhang MD (zhangnianfei9898@163.com)
Savarin Amanda MD (Amanda@163.com)

Version: 7 Date: 16 October 2015

Author's response to reviews: see over
Dear chief editors,

Thank you for your kind letter and the reviewers’ concerning our manuscript entitled “Extracorporeal shock wave therapy in the treatment of bone marrow edema syndrome of the knee: a pilot study (9066061081612224)”. Based on the comments we received, careful modifications have been made to the manuscript. We hope the new manuscript will meet your magazine’s standard. Below you will find our point-by-point responses to the reviewers’ comments/questions:

A. Responses to Comments of Reviewer

I'm afraid we must ask, however, that you also state consent to publish the information contained in the case report. We would suggest including this statement at the end of your case report: "The patient has provided consent to publish the information contained in this case report, as well as figure 5 and the accompanying legend"

**Response:** Thank you very much for your kind review. Based on the comments we received, careful modifications have been made to the manuscript. We hope the new one will meet your magazine’s standard.
We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions. If this article needs further revision, please do not hesitate to contact us.

Sincerely yours,

Fuqiang GAO, Wei SUN,
Department of Orthopedic Surgery,
Beijing Key Laboratory of Arthritic and Rheumatic Diseases,
China-Japan Friendship Hospital, Beijing, 100029, China

Correspondence to: Wei SUN MD. Department of Orthopedic Surgery,
China-Japan Friendship Hospital, Beijing, 100029, China
Tel 86-010-84205009; Email:18901267995@163.com